 CHOHAN /17--Oct- 05

Week 42 starting 17 oct 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  17 Oct 05 :  10/17/2005 11:59:26 AM  
	7
	A1
	MBBR1259
	Prep 2.2 IAP @ warm HF
	295.18

	2
	A2
	MBAR1265
	PT 10 Min. NRJ 11850 A
	4.81

	8
	B1
	SSS654
	Prep 2.2 IAP @ warm HF
	294.73

	11
	B2
	SSS049
	ICS 4 Final connection
	295.36

	9
	C1
	MBAL2192
	ICS 4 Final connection
	293.57

	1
	C2
	MBAR2222
	PT 9 CDAP HV
	1.91

	10
	D1
	SSS292
	ICS 2 Connect Magnet (ICS)
	295.18

	4
	D2
	SSS604
	FS23 MQ Strength @11850
	1.90

	5
	E1
	MBBL3411
	Prep 5 Cool Down
	96.67

	12
	E2
	CRYO
	Cryo leak repair
	295.44

	6
	F1
	MBAR1268
	PT 3 QH Measure
	2.35

	3
	F2
	MBBL2055
	PT 6.3 Training 3
	49.28


  Tempr. from TBSM18 site 
   
 
   
Coordinator this week: Pierre; MTM support: 16-3602

14 complete Magnet Tests cycles  from 11 benches ( E2 still 
under CFB repair)
 9 dipoles + 3  SSS  + 2  SSSS
 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
Situation From Cryogenics:  J-P Lamboy or ACR Rep.
No new magnet tested on E2 since 29 Sept (already 3 weeks !) 

E2 CFB (leak) Repair whole Week this week again?  ( 11 benches usage only)
 Special  Points :
· Special SSS on B1 : connection problems ?
· Special SSS604  on D2 :Great progress thanks to Jackie/Eric/Kevin/Roy shifts
During weekend; SSW was finished Sat . morning. Nothing done since; So how long now for Chaconsa?

· AOB :  Connector for Dipoles when they arrive (Eric/Giorgio) 
[image: image1.png]Magnets under Test from 09-10-2005 to 15-10-2005

- [Bench | Magnet |Shafis| + |Start Date \“qnect (Conling |Cobd Test Way Disconnect | e pate | Total |Signed for
1At MBBL2IE |[v | [07-102005| 27h | 23h | 41h | 15 8h  [12:102008 114 .| Strpping
2 [TBg |mBALI2e9 [X os102005[ 3¢h | 20 | 18h | 13n 7h  [13:102005 s h. | Suipping
3 [Tea2 |mBBRIZET X 03102005 46h | 20 | 7h | 19n | h  [0102005 139 h. | Strpping
4 [TBt [sssess X |® [or-102005| s7h. | 40h | 27h. | 1sh. | 38h  [15102005 209 h. | Stand By
5 [Tee2 [sssieg |X % [os102005| 25h | 13h | 1sh | foh | 21h  [0e-102005 88 . | Stipping
6 [ToB2 [sssais X 0102005 8h | 24h | 16h | 1th | 26h  [15102005 133 h. | Strpping
7 [Tct | mBBRIZES | X 0102005 2¢n | 23h | Wh | 20h | 10h  [11-10:2005 108 h. | Strpping
8 [TBc2 |mBBL2O7I [X 07102005 s4h | 34h | ath | 7h | 13n [14102005 159 h. | Strpping
9 [t01 [sssas X 5102005 71h | 19h | 2¢h | 10h | 19h  [12102005 143 h. | Strpping
101802 [sssew  [X 8092005 46h | 72h | 108h | 10h | sBh  [11-102005 292 h. | Stand By
11 [1BET |MBAR2228 [ X 102005 32h | 49n | sah | t6h | 17h  [15102005 164 h. | Strpping
12 [1BF1 |MBALSBSS [ X os102005 26h | 2h | 20n | 7h | 2h  [10102008 111 b | Stpping
13 (1871 |mBBLItES | X 10102005 29h | 28h | 13h | 17h | tah [14102008 09 h | Spng
141872 |MBARI2E3 [ X 102005 34h | a7h | 23h | 15h | 1th [13102008 130 b | Stpping




 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL2216
	12049.7 A.
(D1-U)
	12794 A.
(D1-L)
	12850 A.
[image: image2.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL1249
	11527.3 A.
(D1-U)
	12625.6 A.
(D1-L)
	12850 A.
[image: image3.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBR1261
	10919.9 A.
(D1-L)
	12070 A.
(D2-L)
	12265.8 A.
(D2-L)
	[image: image4.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS629
	5695 A.
(MQM-I-34)
	5822 A.
[image: image5.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS160
	11838.1 A.
(QE_34)
	12373 A.
(QE_34)
	12500 A.
(QE_12)
	[image: image6.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS349
	11944.9 A.
(QE-12)
	12184.7 A.
(QI_34)
	12834.4 A.
(QI-12)
	[image: image7.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBR1263
	11398 A.
(D1-L)
	11998.9 A.
(D1-L)
	12184.3 A.
(D1-L)
	12458 A.
(D2-U)
	12613.1 A.
(D1-U)
	12737.3 A.
(D1-L)
	12844 A.
(D2_U)
	12850 A.
[image: image8.jpg]



	-
	-
	-
	-
	-

	8
	MBBL2071
	11888.1 A.
(D2-L)
	12347.5 A.
(D1-U)
	12706.7 A.
(D1_U)
	[image: image9.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS249
	11633 A.
(QI-34)
	11949 A.
(QI-34)
	12404 A.
(QI-12)
	12626.2 A.
(QE_34)
	[image: image10.png]



	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS630
	5800 A.
[image: image11.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR2229
	10622.4 A.
(D1-U)
	12286.8 A.
(D1-L)
	12850 A.
[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAL3643
	11088.3 A.
(D2_U)
	11374.1 A.
(D1-U)
	12007 A.
(D1_L)
	12733 A.
(D1_L)
	12850 A.
[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBL1189
	12351.8 A.
(D2_U)
	12512 A.
(D1-U)
	12530.2 A.
(D1-U)
	[image: image14.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAR1253
	11003.7 A.
(D2-L)
	12611.2 A.
(D2-L)
	12768 A.
(D2_U)
	[image: image15.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
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Shows  SSSS work on B1 & D2  & cohabitation with B2 & D1  (arc-SSS) ;  why long disconnection on 1261 A2 ?
[image: image17.png]CH UTILISATION End-AUG-Sept ill 14 Oct 05 ; E? Ciyo problem end Sept: B1 & D2 for IR-SSS ; B? 801 arc-S58

'SMIMS From 15-09-2005 to 14-10-2005

= TBench [ Magnet Name [shafts [Start Date .| Previous [Next | End Date .| Total
I
| OO0
fio[TeB1 sssee [Fale 20002005 | 1560 05-10-2005 [15 days
158[TBB1 [wBacizs0 [False [12-05-2005 on 3 days
{in[T882 [sssis0 [False _|05-10-2005 sh 05-10-2005 | 4 days
118[T882 [sssi6r [False _|01-10-2005 4h 0510-2005 | 4 days
126[T882 _[sssug [False _[27-05-2005 oh 01-10-2005 | 4 days
135[T882 [sseats [False [23-05-2005 h 27-06-2005 | 4 days
fa3[TBB2 [sssone [False [20-05-2005 4h 25:06-2005 | 3 days
151[T882 _[sssaa3 [False 2n 19-09-2005 | 3 days
{60 [TeE2 [MeBL1187 [False 13h S days
{_[TBD1 sssies
75 [TB01 [ssspan [Fabse _|05-10-2005 o 7 days
[17]TB01_[sssor [False _[30-05-2005 ih 0510-2005 | 5 days
{30[TBD1 [sss1s0 [False _[18-05-2005 3n | in [30:052005 12 days
153[TBD1 [ssszas [False _[10-05-2005 in | 3n [18:052005 ] 8 days
{_[TBD2 |sssena 11/10/2005]
77 1802 [sssemn [Fae [26:08.2005 | 43n | sh. |11-10-2005 13 days
fa4[TB02 [sssens [False _|08-05-2005 oh [ e3n [23:00.2005[15 days
| _[TBD2 |sssens 24/08/2005| 08/09/2005 [ 15 days
{12[TBEL |BBL3403 [Fabse _[04-10-2005 in | 2n [08-10-2005 [ 4 days
0[TBEL |BBL2203 [False _|01-10-2005 3n | in [0410-2005 [ 3 days
8[TBEL [wbarsaas [False _[25-05-2005 §h_ | 3n [30-05-2005 [ 5 days
1Bo[TBEL [warsaes [False [21-05-2005 20 | 8n [25:052005 [ 4 days
fa7[TBEL [wBaLizor [False [17-05-2005 6h | 2n [20-05-2005 | 3 days
155[TBEL [wBaciosr [False [13-05-2005 5h_ | 6n [16-05-2005 [ 3 days
{31[TBE2_|wBBL2203 [False _[24-05-2005 6h [131h [20.09-2005] 5 days
G1[TBE2 BB 2001 [Fabse_[18:05-2005 20 | 6n [24:052005] 6 days
[TeE2 JuemLz217 [Fase (13002005 | on_ | 2n [16-092005 [ 5 days





	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_11_A
	TBC2


	Unable to launch hf DAQ during diode test on MB2071.
	AB/CO
	hf
	Guy called AB/CO support. Software bug. A lab-view application vi was still running.

	Tuesday_11_A
	TBC2


	PC fault displayed showing too many valves opened, impeding to launch the training.

Same fault displayed on TBA and TBF
	AB/CO
	hf
	Guy came

PLCCU in TEMA mode and has inhibited the access of the test.

PLCCU put in local mode. This mode is not safe: Warning not connect more that 3 power supplie at the same time. A.Siemko is looking into the problem.

	Wednesday_12_M
	TBE1


	MB2229: HV test cold initial before test, connected to CFB failed at dipole versus ground and quench heater versus ground @1800 V. Rack used was MY92. Test @600V pass OK.
	ICS
	
	Pierre called. IFS box opened. Electric arc happened during a trial.

ICS was asked to remove and change the IFS box,

Problem remained the same with IFS box replaced.

ICS will change the IFS box again.

	Wednesday_12_M
	TBC2


	MB2071: Tau Jump significant seen during training quench.
	MTM
	
	Pierre called.

	Wednesday_12_M
	TBC2


	Interlock on water impeding to launch training quench.
	MTM
	
	Check done: Water flow insufficient in the power supply.



	Wednesday_12_A
	TBF2
	HV test on MB1253 using MY92: the PLC -HV is showing communication error with the server. 
Not enough HV rack available to perform the five HV tests queuing (2 racks are working, MY93 is stopped).
	AB/CO
	HV
	Test OK with MY95.

MY92 was also OK.

2/3 racks available.

	Friday_14_A
	TBC2


	MB2055: errors during IAP @warm on LF heaters discharge. 
Error1: Quench heater tmin H2 and tmin H3 values are high, 1614 and 1642 respectively. 
Error2: Vanaode and Vcathode values both are @ - 1.01.
	MTM
	IAP
	Guy informed.

	Friday_14_A
	TBF2


	IAP at warm: Vanaode and Vcathode values both are @ - 1.01
	MTM
	
	Guy informed. An adaptator need to be installed for Magnet with left polarity since the beginning of the tests.

	Friday_14_A
	TBC2


	MB1278: Quench during SPA test at 12000 A.Power converter tripped.
	AB/PO
	
	

	Friday_14_N
	TBC2


	VSupra Ext and VClamp Ext was permanent trigger on this card..
	ICS
	
	ICS has been informed to check connection.

	Friday_14_N
	TBD2


	Confusion in the injection value for the SSW transfer function measurements
	MTM
	
	Stephane’s comment: The correct value is that of the TO DO LIST i.e. 265 A. This is the new value defined by the MEB in August 2005 for the MQM AT 1.9 K. It replaced the old value of 350 A.

	Saturday_14_N
	TBB1


	SSS654: Superconducting wires for correctors are too short (5cm missing) on magnet side. ICS team can't solder them to CFB side.
	MTM
	
	Guy and TC informed. No need to connect 600 A bus bar for S4 magnets.

	Sunday_15_M
	TBB2


	SSS049: IAP at warm : too low voltage value across the EF circuit i.e. 1 mV instead of 14 mV
	MTM
	
	Guy informed. ICS checked the connection.  Short installed for  previous magnet still installed. OK to proceed with the tests.

	Sunday_15_N
	TBA2


	Shafts 27/28  are scratching during rotation.

MM2 on MB1265: Over-ranges in sector 02/03.
	MTM
	
	Problem solved: Protecting cover is touching the flat cable.

Stephane advised to contact Maryline. Maryline and Fabienne found that the flat cable of the TRU was damaged. Stephane advised to change the TRU.

	Sunday_15_N
	TBB1


	SSS654 : Vclamp and V-Supra : permanent trigger found during IAP@warm.
	ICS
	
	ICS checked the connection again.

600 A bus bar not connected for S4

	Monday_16_M
	TBF2


	MB2055: Jump in voltage going above the threshold.
	MTM
	
	Pierre advised to continue the training.











